 Vc/29-Mar-05
Week 12 starting 21 Mar. 05
SM18 Operation Meeting (Tuesdays)

Bench status on Tuesday 22 morning.
	12
	A1
	MBBL1196
	Prep 1 Lyre Test
	295.18

	2
	A2
	MBBR3275
	PT 6.1 Training 1
	4.55

	3
	B1
	SSS602
	Survey @ Cold SC24
	1.90

	9
	B2
	MBAL1202
	Prep 5 Pumping + Cool Down
	95.7

	10
	C1
	MBAL2155
	Prep 3 CDW1 HV
	294.64

	11
	C2
	MBBL1165
	CR 1 Leak Test
	295.18

	1
	D1
	SSS103
	PT 13.2 Resist. Meas.
	296.43

	6
	D2
	SSS019
	Prep 5 Pumping + Cool Down
	294.73

	8
	E1
	MBBR3294
	PT 12 Warm Up
	298.58

	4
	E2
	MBBR3293
	ICS 6 Disconnect Magnet WP07
	299.2

	5
	F1
	MBAL1203
	PTE Min. NRJ 3KA Diode Test
	1.90

	7
	F2
	MBAL2101
	Prep 5 Pumping + Cool Down
	2.35


· Coordinator this week: Vittorio Remondino; 
· MTM support on call : Maryline Gateau .

· 15  Magnets Tested    :10  dipoles ( 2 with MM, MB3127, MB3284 special test) + 3  SSS ( 1 – Only  HV Test (SSS36), 1- Thermal cycle (SSS129) , 0  with MM) +  1  SSSS (SSS601)
   Disconnected after tests last week ( Sat Midnight to Sat midnight)
· Situation From Cryogenics:  J-P Lamboy or ACR Rep.
· Situation From Mechanics:  J.Halik

· SSS:  N.Smirnov.
-SSW meas. on SSS601.
- 3rd system.
· AOB:  Status of HV rack (Richard), AQA (Marco).
	Magnets under Test from 13-03-2005 to 19-03-2005 

	-
	Bench
	Magnet
	Shafts
	+
	Start Date
	Connect
Time
	Cooling 
Time
	Cold Test 
Time
	Warming 
Time
	Disconnect
Time
	End Date
	Total
	
	

	1
	TBA1
	MBAL3144 
	[image: image1.png]



	


	11-03-2005 
	18 h.
	24 h.
	24 h.
	15 h.
	9 h.
	14-03-2005 
	90 h.
	
	

	2
	TBA2
	MBAR3258 
	[image: image3.png]



	


	13-03-2005 
	20 h.
	24 h.
	23 h.
	23 h.
	12 h.
	18-03-2005 
	102 h.
	
	

	3
	TBA2
	MBBL3127 
	[image: image5.png]



	


	08-03-2005 
	35 h.
	28 h.
	36 h.
	22 h.
	8 h.
	13-03-2005 
	129 h.
	
	

	4
	TBB1
	SSS601 
	[image: image7.png]



	


	28-02-2005 
	62 h.
	20 h.
	158 h.
	30 h.
	60 h.
	14-03-2005 
	330 h.
	
	

	5
	TBB2
	MBBL2143 
	[image: image9.png]



	


	09-03-2005 
	28 h.
	34 h.
	15 h.
	25 h.
	22 h.
	14-03-2005 
	124 h.
	
	

	6
	TBC1
	MBAR3256 
	[image: image11.png]



	


	13-03-2005 
	24 h.
	29 h.
	20 h.
	18 h.
	7 h.
	17-03-2005 
	98 h.
	
	

	7
	TBC1
	MBBL3155 
	[image: image13.png]



	


	09-03-2005 
	21 h.
	26 h.
	15 h.
	15 h.
	7 h.
	13-03-2005 
	84 h.
	
	

	8
	TBC2
	MBAR3297 
	[image: image15.png]



	


	11-03-2005 
	26 h.
	33 h.
	49 h.
	17 h.
	7 h.
	16-03-2005 
	132 h.
	
	

	9
	TBD1
	SSS036 
	[image: image17.png]



	


	09-03-2005 
	60 h.
	57 h.
	52 h.
	16 h.
	18 h.
	18-03-2005 
	203 h.
	
	

	10
	TBD2
	SSS129 
	[image: image19.png]



	


	06-03-2005 
	50 h.
	24 h.
	66 h.
	11 h.
	0 h.
	13-03-2005 
	151 h.
	
	

	11
	TBD2
	SSS129 
	[image: image21.png]



	


	13-03-2005 
	0 h.
	1 h.
	17 h.
	10 h.
	22 h.
	15-03-2005 
	50 h.
	
	

	12
	TBE1
	MBAR3255 
	[image: image23.png]



	


	08-03-2005 
	30 h.
	32 h.
	40 h.
	20 h.
	8 h.
	14-03-2005 
	130 h.
	
	

	13
	TBE1
	MBBR3263 
	[image: image25.png]



	


	14-03-2005 
	38 h.
	31 h.
	17 h.
	15 h.
	12 h.
	19-03-2005 
	113 h.
	
	

	14
	TBE2
	MBAR1046 
	[image: image27.png]



	


	12-03-2005 
	30 h.
	26 h.
	28 h.
	18 h.
	13 h.
	17-03-2005 
	115 h.
	
	

	15
	TBF1
	MBAR3284 
	[image: image29.png]



	


	10-03-2005 
	30 h.
	24 h.
	114 h.
	29 h.
	13 h.
	19-03-2005 
	210 h.
	
	


Quench Information
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Week  13 Mar 05 to 19 Mar 05
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Thermal Cycle

HV Tests @ Cold 
Light green: Connect + Pump down 
Blue :  Wait + Cool down


Ivory: Tests at Cold (36 hr)


Red :  Wait + Warm up (12 hr)
    Dark Green: Disconnect (8 hr)
	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Tuesday _15_M


	TBF1
	Air leak in the CFB: quench recovery difficult.


	ACR
	
	J.P Lamboy managed to solved the problem: re-start of the  extended test ½ day after.



	Tuesday _15_A


	TBC1
	communication problem between local sun server and PLC


	AB/CO?
	
	OK after restarting server process on local sun station downstairs

	Tuesday _15_A


	TBA1
	A1 anticryostat problem 
	ICS
	
	Solved by Salem: bad connections

	Tuesday _15_A


	all
	unable to log in to HV Test application (error message: "The user name does not exist") 
	AB/CO
	
	M.Marchesotti solved the problem: The user server at 17.30 started distributing an empty Users.cfg file. User server stopped and the configuration files were distributed manually to all the workstations. 

	Tuesday _15_N


	TBF1
	Error at CPU adress and Acquisition trigger not received Error on motor unit 2...Test Aborted during the cycle..
	MTM
	
	Salem called. Intervention of Frederic and Patrick. The end stop detection was not working properly in aperture 2. The shaft went into the mechanical end stop and the clutch was slipping.

	Thursday _17_M

 Thursday_17_A
Thursday_17_N
	TBD1

TBA1
	XMT.MY 95 HV rack connector is faulty (passive/active bus bar)
XMT.MY 93 HV rack did not work

The magnet name was not correct.

XMT.MY 92 HV rack did not work
The magnet name through bench selection is not working


	ICS
	
	Repair is needed. Done by Fabienne Friday.

Changed over to XMT.MY95.
Name entered manually.
 

	Tuesday _15_N


	TBD1
	WP17 : launched using XMT.MY 95 HV (authorisation of Salem): wrong polarity in the setting
	MTM
	
	Salem asked to lunch the test with the repeat option..

	Tuesday _15_N


	TBF1
	Rocla did not work: it impeded ICS team to remove the shafts from the 3284 magnet
	ICS
	
	Shift leader asked to cryo team to stop the warm up. TC informed. 

TC asked the cryo team to put at least the magnet in stand by.  

Comment: if the magnet cannot be put in stand by do not cool it down again : the shafts will be removed at warm by MTM mechanics.

	Tuesday _15_N


	TBD2
	SSS128: Problem in the extension number of training quenches: a02 for quench1 and a04 for quench 2.


	MTM
	
	

	Friday _16_M


	TBD2
	SSS 128: Config Error when launching next training quench ´Error in DB while executing hr_confgen´
	MTM
	
	Salem informed. 



	Friday _16_M


	TBA1
	IAP at cold  repeated on 1184 magnet as it was carried earlier with Q-Loc cable connected for LQA & not for shaft..
	MTM
	
	Salem connected the right cable for long shaft and test was repeated 



	Friday _16_M


	TBA1
	Magnet quenched at 12681 A.

AQA analysis of file .zb01 shows an alert that QH heaters were not charged though they were charged and discharged also as a result of this quench.
	MTM
	
	











